Author’s response to reviews

Title: Stress Induced Hypertensive Response : Should it be evaluated more carefully?

Authors:

Nagehan Nagehan Kucukler (drnagehan@gmail.com)
Fatih Yalcin (fyalcin1@jhmi.edu)
Theodore P Abraham (tabraha3@jhmi.edu)
Mario J Garcia (mariogar@montefiore.org)

Version: 2 Date: 8 May 2011

Author’s response to reviews:

We are submitting the review, entitled “Stress Induced Hypertensive Response : Should it be evaluated more carefully?” for evaluation in the “Cardiovascular Ultrasound”.

We state that all authors have read and approved submission of the review that has not been published and is not being considered for publication elsewhere.

Please consider the review for publication in the “Cardiovascular Ultrasound”.

Regards